# UC Santa Cruz UC Santa Cruz Previously Published Works

## Title

Plant Uptake of Atmospheric Carbonyl Sulfide in Coast Redwood Forests

**Permalink** https://escholarship.org/uc/item/0s55h5nk

**Journal** Journal of Geophysical Research Biogeosciences, 122(12)

**ISSN** 2169-8953

## Authors

Campbell, JE Whelan, ME Berry, JA <u>et al.</u>

**Publication Date** 

2017-12-01

## DOI

10.1002/2016jg003703

Peer reviewed

# **@AGU**PUBLICATIONS

# Journal of Geophysical Research: Biogeosciences

## **RESEARCH ARTICLE**

10.1002/2016JG003703

#### **Key Points:**

- First atmospheric carbonyl sulfide (COS) measurements in a coast redwood forest
- Measurement shows persistent daytime COS depletion and COS:CO<sub>2</sub> linear relationship
- Results support use of COS as a biogeochemical tracer of photosynthesis

Supporting Information:

Supporting Information S1

Correspondence to: J. E. Campbell, elliott.campbell@ucsc.edu

#### Citation:

Campbell, J. E., Whelan, M. E., Berry, J. A., Hilton, T. W., Zumkehr, A., Stinecipher, J., ... Loik, M. E. (2017). Plant uptake of atmospheric carbonyl sulfide in coast redwood forests. *Journal of Geophysical Research: Biogeosciences*, 122. https:// doi.org/10.1002/2016JG003703

Received 3 NOV 2016 Accepted 26 NOV 2017 Accepted article online 13 DEC 2017

## Plant Uptake of Atmospheric Carbonyl Sulfide in Coast Redwood Forests

J. E. Campbell<sup>1</sup> <sup>[1]</sup>, M. E. Whelan<sup>2</sup> <sup>[1]</sup>, J. A. Berry<sup>3</sup>, T. W. Hilton<sup>2</sup> <sup>[1]</sup>, A. Zumkehr<sup>2</sup> <sup>[1]</sup>, J. Stinecipher<sup>2,4</sup> <sup>[1]</sup>, Y. Lu<sup>2</sup>, A. Kornfeld<sup>3</sup>, U. Seibt<sup>5</sup>, T. E. Dawson<sup>6</sup>, S. A. Montzka<sup>7</sup> <sup>[1]</sup>, I. T. Baker<sup>8</sup> <sup>[1]</sup>, S. Kulkarni<sup>9</sup>, Y. Wang<sup>10</sup> <sup>[1]</sup>, S. C. Herndon<sup>11</sup> <sup>[1]</sup>, M. S. Zahniser<sup>11</sup>, R. Commane<sup>12</sup> <sup>[1]</sup>, and M. E. Loik<sup>1</sup> <sup>[1]</sup>

JGR

<sup>1</sup>Environmental Studies Department, University of California, Santa Cruz, CA, USA, <sup>2</sup>Sierra Nevada Research Institute, University of California, Merced, CA, USA, <sup>3</sup>Department of Global Ecology, Carnegie Institution, Stanford, CA, USA, <sup>4</sup>Lawrence Livermore National Laboratory, Livermore, CA, USA, <sup>5</sup>Department of Atmospheric and Oceanic Sciences, University of California, Los Angeles, CA, USA, <sup>6</sup>Department of Integrative Biology, University of California, Berkeley, CA, USA, <sup>7</sup>Global Monitoring Division, NOAA Earth System Research Laboratory, Boulder, CO, USA, <sup>8</sup>Department of Atmospheric Science, Colorado State University, Fort Collins, CO, USA, <sup>9</sup>California Air Resources Board, Sacramento, CA, USA, <sup>10</sup>Department of Atmospheric Science, University of Bremen, Bremen, Germany, <sup>11</sup>Aerodyne Research, Inc., Billerica, MA, USA, <sup>12</sup>School of Engineering and Applied Sciences, Harvard, Cambridge, MA, USA

**Abstract** The future resilience of coast redwoods (*Sequoia sempervirens*) is now of critical concern due to the detection of a 33% decline in California coastal fog over the 20th century. However, ecosystem-scale measurements of photosynthesis and stomatal conductance are challenging in coast redwood forests, making it difficult to anticipate the impacts of future changes in fog. To address this methodological problem, we explore coastal variations in atmospheric carbonyl sulfide (COS or OCS), which could potentially be used as a tracer of these ecosystem processes. We conducted atmospheric flask campaigns in coast redwood sites, sampling at surface heights and in the canopy (~70 m), at the University of California Landels-Hill Big Creek Reserve and Big Basin State Park. We simulated COS atmosphere-biosphere exchange with a high-resolution 3-D model to interpret these data. Flask measurements indicated a persistent daytime drawdown between the coast and the downwind forest (45 ± 6 ppt COS) that is consistent with the expected relationship between COS plant uptake, stomatal conductance, and gross primary production. Other sources and sinks of COS that could introduce noise to the COS tracer technique (soils, anthropogenic activity, nocturnal plant uptake, and surface hydrolysis on leaves) are likely to be small relative to daytime COS plant uptake. These results suggest that COS measurements may be useful for making ecosystem-scale estimates of carbon, water, and energy exchange in coast redwood forests.

**Plain Language Summary** The future resilience of coast redwoods (*Sequoia sempervirens*) is now of critical concern due to the detection of a 33% decline in coastal fog that sustains these forests. However, monitoring the potential impacts on redwood forests is challenging because the unique features of the coastal environment interfere with ecological measurement techniques. Here we propose a solution involving a new technique using carbonyl sulfide, an atmospheric chemical that is related to carbon dioxide. We found that the redwoods remove carbonyl sulfide from the atmosphere, which is a critical prerequisite for using this technique to explore redwood resiliency.

#### 1. Introduction

Coast redwoods (*Sequoia sempervirens*) uniquely occur in the California Current upwelling region, suggesting a reliance on coastal fog conditions (Figure 1). Climatological data show a strong statistical relationship between fog and transpiration in coast redwoods (Johnstone & Dawson, 2010). These coastal data also reveal a 33% decline in coastal fog over the 20th century (Johnstone & Dawson, 2010).

The coastal fog decline is of particular concern for redwood forests for multiple reasons. First, coast redwoods are thought to be vulnerable to future changes in fog because they may be poor regulators of water use and may rely on the presence of fog during dry summer months to increase water uptake and reduce evapotranspiration (Dawson, 1998; Simonin et al., 2009). Second, the enormous stature of coast redwoods results in significant interception of fog water, causing fog drip that not only supports coast redwoods but also enhances soil moisture and streamflow for the benefit of coastal ecosystems (Sawaske & Freyberg, 2015). Third, coast

©2017. American Geophysical Union. All Rights Reserved.



**Figure 1.** Coastal California domain of upwelling, fog, and coast redwoods. (a) Sea surface temperature from MODIS remote sensing data delineate the coastal upwelling zone of low temperatures(June 2013), (b) summer mean coastal fog frequencies (Johnstone & Dawson, 2010), and (c) coast redwood range along the U.S. Pacific Coast and location of COS measurements (diamond, Sutro; triangle, Big Basin; circle, R/V *Atlantis*; square, Big Creek).

redwoods may play a critical role in coupling the coastal ocean-atmosphere-land system as suggested by more general theoretical work showing positive feedbacks between coastal upwelling and terrestrial ecosystems (Diffenbaugh et al., 2004).

While much has been learned about coast redwood ecophysiology at the spatial scale of leaves and individual trees (Ambrose et al., 2009; Burgess & Dawson, 2004; Dawson, 1998; Ewing et al., 2009; Johnstone et al., 2013; Limm et al., 2009; Simonin et al., 2009), we lack measurements at larger spatial scales that are needed for a robust understanding of the interactions between coast redwoods and fog (Fernández et al., 2015; Potter, 2012). In particular, ecosystem-scale measurements of gross primary production (GPP) and stomatal conductance are challenging in redwood forests. GPP estimates using eddy covariance CO<sub>2</sub> measurements face logistical and theoretical challenges due to the complex topography and unstable terrain of redwood forests. Alternatively, regional variation in atmospheric CO<sub>2</sub> cannot be used to infer regional GPP because atmospheric mixing at the regional scale prevents analytical partitioning of the co-located GPP and respiration signals (Campbell et al., 2008). Furthermore, stomatal conductance estimates that are typically based on eddy covariance water vapor measurements are hampered by the presence of coastal fog.

These methodological challenges could potentially be addressed using measurements of atmospheric carbonyl sulfide (COS or OCS). COS is a trace gas that is removed from the atmosphere when it diffuses into plant leaves through stomata and is irreversibly hydrolyzed inside the leaf by the photosynthetic enzyme carbonic anhydrase (Berry et al., 2013; J. Campbell, Kesselmeier, et al., 2017; Campbell et al., 2008; Montzka et al., 2007; Sandoval-Soto et al., 2005; Stimler et al., 2010). The global COS plant sink is largely balanced by COS sources from oceans (Launois et al., 2015) and Asian industry (Campbell et al., 2015). This spatial separation of the dominant source and sink suggests that observed variations in atmospheric COS over terrestrial ecosystems should provide a tracer of plant physiology processes. While other sources and sinks can be large at some points in time and confound the COS approach (Commane et al., 2015; Maseyk et al., 2014), the regional dominance of COS plant uptake over continents during the growing season has led to the use of this tracer for estimating GPP and stomatal conductance at ecosystem to global scales (Asaf et al., 2013; J. E. Campbell, Berry, et al., 2017; Hilton et al., 2017; Wehr et al., 2017).

Unlike CO<sub>2</sub>, the land-atmosphere exchange of COS during the growing season is dominated by the plant flux in most ecosystems and other ecosystem sources and sinks are generally less significant. Thus, while regional CO<sub>2</sub> measurements are useful for estimating net ecosystem exchange, COS measurements could be applied to estimate regional GPP and stomatal conductance. Furthermore, while atmospheric water vapor concentrations are influenced by sources from plants and fog, atmospheric COS mixing ratios are more generally

Summary of Field Campaigns and Sampling Sites						
Campaign	Date	Sample site	Location	Sample elevations (m above surface)		
Big Creek 1	24–25 June 2015					
		Coast	36°4′10.10″N, 121°35′54.97″W	2 m		
		Redwood	36°3′58.56″N, 121°33′16.42″W	2 m		
Big Creek 2	12–13 December 2015					
		Coast	36°2′39.75″N, 121°35′7.37″W	2 m		
		Redwood	36°3′23.65″N 121°32′27.93″W	4 m, 34.8 m, 70.5 m		
Big Basin	26–27 May 2015					
		Coast	37°05′51.0″N, 122°16′51.1″W	2 m		
		Redwood	37°12′46.0″N, 122°13′14.9″W	2 m		
		Grassland	37°18′56.7″N, 122°18′23.1″W	2 m		

controlled by the COS plant sink (see section 4). In order to explore the potential for the COS approach for coast redwoods, we conducted the first ambient COS observations in Northern California forests. We also collected atmospheric samples from sites adjacent to the redwoods to observe marine boundary layer (coastal) and grassland influences on atmospheric COS. The flask samples were analyzed for COS and CO<sub>2</sub> mixing ratios using a quantum cascade laser spectrometer referenced to the NOAA gas standard. To interpret these measurements, we simulated atmospheric mixing ratios using a high-resolution 3-D model of atmosphere-biosphere exchange and transport. We also used the model to interpret atmospheric COS measurements from a previously conducted survey of coastal California measurements of COS and CO<sub>2</sub> aboard the R/V *Atlantis* during the California Nexus Experiment (CalNex) and NOAA global air-monitoring data from Sutro Tower (San Francisco, CA, USA). We discuss the results with respect to the potential use of COS as a GPP and stomatal conductance tracer for coastal redwoods. Extended methods and results are in the supporting information (Bloem et al., 2012; Li et al., 2007; Vacher et al., 2016).

#### 2. Methods

#### 2.1. Flask Sampling

We conducted three flask sampling campaigns at coast redwood sites in the University of California Natural Reserve System Landels-Hill Big Creek Reserve (Monterey County, CA) and Big Basin State Park (Santa Cruz County, CA). These campaigns included sampling in a Lagrangian framework (upwind coast and downwind forest sites), over diurnal cycles, at a gradient of elevations (near the land surface through near the top of the redwood canopy), and in the summer and winter as summarized in Table 1. During all three campaigns, fog was present in the early morning while the daytime was mostly clear.

These Lagrangian campaigns included sampling in redwood canopy sites and adjacent coast sites, while one of these campaigns included sampling in an adjacent grassland site. The coast sites were within 100 m of the ocean, while the redwood sites were approximately 10 km inland. The predominant northwesterly winds generally created conditions where the coast sites sampled well-mixed, incoming marine background air while the downwind redwood sites sampled continental air. However, other transport regimes were also apparent in the data and are explored in the results. The duration of each campaign was approximately 24 h with a sample collected every 30 min to 2 h.

Sources and sinks in the sampling materials are a known challenge for COS measurements. For example, certain plastics in soil chambers have been shown to create a temperature-dependent source (Maseyk et al., 2014). Here we only used sampling equipment that has been demonstrated in previous work to be suitable for COS analysis, including Synflex tubing and flask packages that have been extensively evaluated through field experiments and the NOAA global air-monitoring network (Montzka et al., 2007; Sun et al., 2016).

Ambient air was collected with an automated sampling system consisting of programmable compressor packages (PCPs) and programmable flask packages (PFPs) (High Precision Devices, Inc., Boulder, CO). Software embedded in the PFP automated gas sampling for the diurnal campaigns with minimal intervention, for example, replumbing gas lines for sampling at different canopy heights. Each PFP contained 12 individual 0.7 L glass flasks with elastomer-free glass stoppers on an inlet and outlet. The flasks were first prepared by flushing with nitrogen gas at 15 L min<sup>-1</sup> for 1 h to prevent contamination and to allow us to

Table 1

detect, postcampaign, flasks that did not fill in the field as programmed. In the field, at preprogrammed intervals, the PCP compressor flushed the external and internal lines with ambient air for a given number of minutes depending on the volume of the sampling line. The sample flask was similarly flushed and then filled to 200–275 kPa. This compressor was powered by an external 12 V battery.

The sampling lines consisted of Synflex tubing that was extended either near the ground surface or into the redwood canopy (Table 1). Particulate matter was excluded using stainless steel filters (2–7  $\mu$ m), and the lines extended into the canopy had aluminum shields to prevent rain entrainment.

For canopy air sampling, this study adopted a "trees-not-towers" approach, a lower cost method of obtaining samples from different heights within the canopy. In lieu of depending on human-made towers, we were able to sample air in remote locations high in the canopy by climbing the trees and installing sampling lines at various heights. The lines were tied to the tree branches with weatherproof rope and were supported by an aluminum bracket that allowed the gas inlet to be extended away from the tree trunk. This method has the added advantage of avoiding measuring disturbance by other tower instruments, which may be a problem at many sites since COS is outgassed by many common plastics.

#### 2.2. Trace Gas Measurement

Flask samples were analyzed in the laboratory using a quantum cascade laser developed to simultaneously measure COS with parts-per-trillion (ppt) precision and CO<sub>2</sub> to a resolution of 0.1 ppm (Aerodyne Inc., Billerica, Massachusetts; Stimler et al., 2009). The standard system was designed to measure COS on a continuous flow basis. However, to increase the number of replicate measurements that could be performed on a single flask, we developed an automated system that operates on a closed measurement cell (i.e., no flow). We included a magnesium perchlorate water trap upstream of the sample cell to reduce spectral noise introduced in humid samples (Kooijmans et al., 2016). All sample measurements were calibrated by alternating sample measurements with measurements of a secondary standard calibrated to a NOAA certified standard. The uncertainty, taken to be the standard deviation of three repeated measurements on the same flask, was on average ~7 ppt.

#### 2.3. Land-Atmosphere Flux Modeling

We created a high-resolution, atmosphere-biosphere surface flux estimate of COS plant uptake with an hourly, 1 km resolution covering our regional domain. We generated this high-resolution estimate by down-scaling our previous global simulations of COS plant uptake with the carbon cycle model SiB3, which is detailed in Berry et al. (2013) and summarized as follows.

The SiB3 model simulates the diffusion of atmospheric COS into leaves based on resistances to diffusion imposed by the stomatal pore and leaf boundary layer (conductance = 1/resistance) using an empirical relationship (Ball et al., 1987; Seibt et al., 2010; Stimler et al., 2010). Once COS has diffused into the leaf, it is irreversibly hydrolyzed at a rate proportional to the COS partial pressure in the chloroplast. The activity rate for this reaction is modeled as a linear function of photosynthetic capacity ( $V_{max}$ ) with the proportionality constant estimated from leaf gas exchange measurements (Stimler et al., 2010; Stimler et al., 2012).

While the SiB3 model of COS plant uptake has been validated at the global scale using satellite and airmonitoring observations (Berry et al., 2013; Glatthor et al., 2015; Hilton et al., 2015; Kuai et al., 2015; Wang et al., 2016), the SiB3 spatial resolution (1°) is too coarse to resolve regional features such as the coastal forests (Figure S1 in the supporting information). To spatially downscale this global flux we used Moderate Resolution Imaging Spectroradiometer (MODIS) monthly leaf area index (LAI) data (Knyazikhin et al., 1999) at 1 km resolution as a spatial proxy to spatially redistribute the global flux at each hourly time step of the global flux data. We used the global flux to set the sum of the COS flux over the entire model domain, and we used the MODIS LAI to set the spatial distribution of this flux.

#### 2.4. Additional Sources and Sinks

In addition to plant uptake, our simulations also considered the potential for alternative sources and sinks to influence atmospheric COS mixing ratios. Here we used previously reported gridded inventories for anthropogenic emissions and plant uptake (Kettle et al., 2002). Atmospheric analysis suggests that anthropogenic sources in the U.S. are small compared to COS plant uptake (Blake et al., 2008; Campbell et al., 2015; Guo et al., 2010; Zumkehr et al., 2017), suggesting that this inventory could provide a conservative estimate of

the anthropogenic signal. While field soil chamber observations show that some soils can switch from a sink to a source at high temperature (Maseyk et al., 2014); lab incubations of soils show a small soil flux in forests that is generally consistent with gridded inventories (Whelan et al., 2016). Biomass burning is another significant global source (Campbell et al., 2015). Although there were fires in our study region in the summer of 2015, there were no significant burning events during the days that we collected air samples. Oceans are thought to be the dominant global source but are not modeled directly in these simulations because our analysis limits the ocean influence by focusing on the difference between the upwind coastal site measurements and downwind redwood site measurements.

#### 2.5. Chemical Atmospheric Transport Modeling

We simulated 3-D, time-varying atmospheric COS mixing ratios using an atmospheric chemical transport model. The transport model is the Sulfur Transport and dEposition Model (STEM) that has been widely applied for analysis of atmosphere-biosphere interactions for the U.S. and other regional domains (Campbell et al., 2007, 2008; Hilton et al., 2015; Sandu et al., 2005). Input to the model includes gridded surface fluxes (sections 2.3 and 2.4) and 3-D meteorological data (see below). STEM is a mesoscale, 3-D Eulerian atmospheric chemical transport model employing a finite differences numerical approach to solve the chemical continuity equation.

Simulations were completed for the field sampling campaign periods including 1 week of spin-up. Constant COS mixing ratio boundary conditions were used to assess the individual contributions of the different sources and sinks to atmospheric COS variation. In addition to constant boundary condition, we also explored the effect of temporal and spatial variations in boundary conditions using output from global chemical transport simulations for COS from the GEOS-Chem model (Wang et al., 2016). Model results were extracted for the times and locations of observations (all observations were within the lowest model vertical layer).

We generated meteorological input data for STEM using the Weather Research and Forecasting version 3.7 (WRF) model. Our WRF runs used a series of nested domains with 36 km, 12 km, and 4 km horizontal resolutions to examine state-level trends as well as a Northern California centered domain with nested 9 km, 3 km, and 1 km subdomains for a detailed investigation of the flask measurements (Figure S2). The STEM simulations used the innermost domain from the WRF runs as meteorological input. We provided boundary conditions to the WRF runs using National Centers for Environmental Prediction final global tropospheric analyses (1° resolution) and WRF options that were consistent with previous WRF studies of coastal California (Angevine et al., 2012).

#### 2.6. Supportive Coastal Observations

We used our field measurements and high-resolution simulations to interpret two additional sets of regional data: CalNex cruise and Sutro Tower air-monitoring data. A COS laser spectrometer was deployed on the R/V Atlantis ship during the CalNex experiment in May and June of 2010 (Commane et al., 2013). The R/V Atlantis data had a precision of 2 ppt for 50 s averages. The absolute spectroscopic accuracy, determined from analysis of NOAA calibrated standard gas on the ship, was ~5% of the total OCS (~20 pptv). During the morning of 2 June offshore of Santa Cruz, CA (approximately 36°53′60″N, 122°0′36″W), the R/V Atlantis data revealed a negative correlation between observed COS and CO<sub>2</sub>, and simulated back trajectories suggested the potential for a nocturnal COS sink in the coastal terrestrial ecosystems. The magnitude of this nocturnal sink in relation to the GPP-related daytime sink is unknown. While these offshore measurements were too far from the land to characterize specific ecosystems, they were downwind of a wide expanse of coast redwoods.

Long-term flask observations of atmospheric COS at Sutro Tower (37°45′18.72″N, 122°27′10.08″W) were collected every 12 h as part of the NOAA global air-monitoring network (Montzka et al., 2007). Flask samples are shipped to Boulder (CO) for analysis by gas chromatography and mass spectrometry. The median replicate precision for COS flask samples from the NOAA network is 0.4% (~2 ppt), and 95% of all pairs have a replicate precision of less than 1.3% (<6.3 ppt). These observations are designed to characterize broad trends associated with the free troposphere and could provide useful information on the background COS trends for coastal COS experiments.



**Figure 2.** Measurements of COS and CO<sub>2</sub> in ambient air from three field campaigns (Big Basin, Big Creek 1, and Big Creek 2) at coast redwood and adjacent sites. All sample heights are near-surface (2–4 m aboveground level) except for a subset of the forest site samples during the Big Creek 2 campaign, which included samples at 70 m and 35 m aboveground level. Measurement uncertainties are the standard deviation of multiple measurements on each flask; many symbols obscure the error bars. Site locations in Figure 1c and Table 1. The gray shading is nighttime.

#### 3. Results

#### 3.1. Flask Observations

The field campaigns provide a first view into COS variation in coast redwood forests (Figure 2). COS atmospheric mole fractions at the surface (2 m aboveground level) at the redwood sites are consistently lower than upwind coastal sites for all daytime measurements in the two summer campaigns (Big Basin and Big Creek 1). This persistent horizontal COS drawdown between the coast and the downwind forest site is consistent with a large COS sink in the coast redwoods.

The relatively high temporal resolution measurements during the Big Basin campaign yielded a  $45 \pm 6$  ppt (mean  $\pm$  SE) horizontal drawdown between the coast and redwood sites during the daytime onshore flow period (Figure 2, left column). The Big Creek 1 campaign had only two observations available for calculating the difference between the redwood and upwind coast sites during the day, but these drawdown measurements were comparable to the Big Basin campaign (42 ppt and 68 ppt; Figure 2, middle column).

While COS mixing ratios during the summer campaigns at the redwood site were significantly lower than the coastal site, the grassland site had similar COS mixing ratios to the upwind coastal site (Figure 2, yellow squares). Our grassland measurements were collected during the end of May, which is climatologically in the dry season but was exceptionally dry during the drought of 2015 (Figure S9). The relatively insignificant depletion of COS over the grassland at the end of May was consistent with the expected low GPP rates and low COS plant uptake previously observed at a coastal grassland field study in this region, which was dormant or senescent (Whelan & Rhew, 2016).



Figure 3. Correlation of COS and CO<sub>2</sub> observations for (a) daytime and (b and c) nighttime ambient samples at coast redwood forest sites (triangles) and on the beach adjacent to the forest (circles). The nighttime coast observations occurred during a period of coastal land breezes. The daytime forest observations occurred during onshore flow. Linear fits are summarized in Table 2.

Diurnal COS trends appear to be associated with the interaction of the land surface flux and atmospheric mixing events. At the Big Basin forest site, the COS minima and CO<sub>2</sub> maxima during the evening and morning coincided with periods of decreased continental boundary layer heights (Figure S6). The correlation between COS and the boundary layer height is consistent with the land surface being a net COS sink. The pulse of low COS air observed at the Big Basin coast site in the evening may be due to katabatic winds that transport continental air to the coast as suggested by the observed shift in wind (Figure S7) and the similarity between the COS:CO<sub>2</sub> correlation at forest and coastal sites described below (Figure 3).

A strong positive correlation between daytime observed COS and CO<sub>2</sub> mole fractions was seen in the relatively high temporal resolution Big Basin measurements (Figure 3a). Similar relationships have also been observed in large-scale observations from continental airborne campaigns (Campbell et al., 2008) and global air-monitoring data (Montzka et al., 2007) (Table 2). The correlation is roughly consistent with the ratio of GPP and net ecosystem CO<sub>2</sub> exchange (NEE) and the regional dominance of COS daytime plant uptake over other terrestrial sources and sinks of COS (Campbell et al., 2008).

While the daytime COS and CO<sub>2</sub> data had a positive correlation, the nocturnal data from the Big Basin forest site were negatively correlated (Figures 3b and 3c). The negative relationship is consistent with a nocturnal ecosystem sink of COS due to nocturnal stomatal conductance and soil uptake and CO<sub>2</sub> source from respiration. Previous work has shown incomplete stomatal closure in redwoods at night (Dawson, 1998), which should lead to a small nocturnal COS plant sink. Nocturnal COS sinks from plants and soils have been observed in other ecosystems to be on the order of 10 to 20% of daily COS uptake (Commane et al., 2015; Kooijmans et al., 2017; Maseyk et al., 2014).

Observed Correlation of Atmospheric COS and $CO_2$ in Daytime and Nighttime Samples							
Time	Reference	Site	Slope (ppt COS/ppm CO <sub>2</sub> )	R <sup>2</sup>			
Daytime							
	This study	Big Basin/forest	4.4	0.8			
	Campbell et al. (2008)	Continental airborne	5.7	NA			
	Montzka et al. (2007)	Northern Hemisphere surface	6	0.9			
Nighttime							
-	This study	Big Basin/forest	-2.4	0.7			
	This study	Big Basin/coast	-3.2	0.9			
	This study	Big Creek 2/forest	-3.2	0.9			
	This study	Big Creek 2/coast	-2.8	0.7			
	Commane et al. (2013)	R/V Atlantis/Monterey Bay	-3.1	0.9			

#### Table 2

A similar negative COS to  $CO_2$  correlation during the night was also observed at the coastal sites in the Big Basin and Big Creek campaigns. The consistency between the regression slope of the nocturnal measurements from the coastal and inland measurements suggests that a land-breeze may have caused an outflow of continental air to influence the coastal site. Wind observations during the sampling campaigns also show the presence of a land-breeze during the nights and early mornings (Figure S7). Evidence of land-breezes in COS data are also apparent in the ship-based CalNex data (Commane et al., 2013).

No significant daytime correlations of COS and  $CO_2$  were observed from the Big Creek 1 data. This may be due to the limited number of samples collected during this campaign.

Even excluding nocturnal land-breeze periods, significant COS variation was observed during the summer field campaigns in the upwind coastal sites. The COS variation was approximately 50 ppt. These changes in background mixing ratios may be associated with synoptic events (see section 3.3). The significant variation in the coastal measurements suggests that the use of COS as a biogeochemical tracer for coastal systems will require quantification of background variability at a subdaily time resolution.

While the two summer campaigns (Big Creek 1 and Big Basin) were subject to northwesterly flow, the winter sampling during Big Creek 2 was subject to northeasterly flow creating offshore wind conditions. A vertical profile during this campaign was sampled in the early morning when the nocturnal boundary layer was likely still intact. The vertical gradient observed at this time may be due to the effect of a shallow nocturnal boundary layer.

#### 3.2. Atmosphere-Biosphere Simulations

High-resolution simulations of surface fluxes and chemical atmospheric transport were used to interpret the field observations. Atmospheric simulations with the plant uptake, soil uptake, and anthropogenic emissions provide information on the contribution of these alternative components of the regional budget. Simulations show significant reduction in COS mixing ratios over forests due to plant uptake of COS but only small effects on mixing ratios from soil and anthropogenic fluxes (Figure 4).

The relatively high temporal resolution data from the Big Basin campaign provide a point of comparison for these simulations. The horizontal gradients from the inland sites to the coast site show strong similarities between the observations and the simulated mixing ratios extracted for the times and locations of the flask samples (Figure 5). The consistency between the model and measurements suggests that plant uptake may be the dominant driver of the coastal COS drawdown. While future development of modeled canopy mixing and vegetation flux will be needed to infer fluxes from COS measurements, our preliminary simulations suggest that the model can roughly approximate the horizontal spatial trends in the observations.

We also compared our simulations to the R/V Atlantis observations of low COS mixing ratios on 2 June in the northern Monterey Bay region. While the consistency between observed COS:CO<sub>2</sub> correlation for the R/V Atlantis and our redwood flask measurements provides strong evidence of a forest influence, we find further support of this hypothesis in our model results. The simulations show a coastal land-breeze advecting the redwood air out to sea and cyclonic circulation trapping this air mass in the region of the ship (Figure 6a). This circulation, named the Santa Cruz eddy (SCE), is a climatological feature in this region that occurs in 80% of summer days due to the prevailing northwesterly flow interacting with the Santa Cruz Mountains (Archer et al., 2005; Tseng et al., 2012).

Although the timing of the simulated drawdown during this land-breeze event matches the observations, the simulated drawdown was roughly half the magnitude of the observed drawdown (Figure 6b). The soil sink is unlikely to explain this discrepancy due to its small magnitude. Alternative simulations with the plant sink doubled and the simulated transport resolution increased to 1 km did not improve the model-observation agreement. Uncertainty in the 3-D transport is a likely source of error as the model shows a roughly 5 km horizontal offset in the location of a 30 ppt depletion that is consistent with the magnitude of the observed depletion (also see section S3 in the supporting information).

Two additional STEM modeling sensitivity experiments were run by adjusting the GPP input by 50% (Figure S10). This is a conservative estimate of GPP uncertainty in the region given that North American continental GPP uncertainty is approximately 50% (Huntzinger et al., 2012). The GPP uncertainty propagates to a 30 ppt drawdown uncertainty, which is supportive of the approach because it is considerably larger than the



**Figure 4.** Atmospheric COS mixing ratio drawdown from 3-D regional simulations. (a) Map of midday COS vertical drawdown due to simulated plant uptake and location of three Big Basin field campaign sites. Vertical drawdown is the difference between the COS mixing ratio in the free troposphere and the mixing ratio in the surface model layer. (b) Simulations of average continental COS drawdown from simulations driven by plant uptake (green), anthropogenic emissions (red), and soil uptake (brown).

7 ppt measurement uncertainty. Such sensitivity simulations may also be used to improve the site selection for future work. For example, the region of highest sensitivity was generally to the southeast of the forest field site used in the Big Basin campaign.

#### 3.3. Background Variation

Inferring ecosystem processes from ambient COS mixing ratios at a redwood site will require knowledge of the background variation. During onshore flow conditions, the background signal is associated with the inflowing air to a site, which may be influenced by more distant sources and sinks. Our field campaigns utilized multiple approaches to assess the background variation. First, we obtained coast site measurements that were generally upwind of the forested sites. This is supported by the agreement between the horizontal



**Figure 5.** Comparison of simulated (red) and observed (gray) daytime COS drawdown during the Big Basin field campaign. Drawdown is the difference between the inland sites and the coast site. The two inland sites were in the coast redwoods and a coastal grassland. The error bars are standard error of the sample-mean difference.

gradients in the modeled and observed data (Figure 5). Below we also consider the potential for regional tall tower observations and global chemical transport simulations to provide information on background variability.

We compare our coastal observations to ongoing NOAA measurements made at Sutro Tower. Sutro Tower is a coastal observatory within 150 km of our field campaign sites. The Sutro observations have a day to day variation that is roughly consistent with the 50 ppt variation at our coastal sites (Figure 7). Consistent calibration approaches were used in our methods and at the NOAA laboratory. While the data are roughly consistent, some differences could be explained by the large distance between Sutro Tower and our sites.

The R/V Atlantis data also show some similarities to Sutro data (Figure 8a). A 50 ppt increase in COS mixing ratios in the R/V record beginning on 16 May, when the ship was still in the Los Angeles basin, was thought to be due to a shift from continental influence to marine influence (Commane et al., 2013). This feature is also observed in the Sutro record, which is consistent with the large spatial extent of background variation. This consistency suggests that Sutro observations may be useful for obtaining background information across the full extent of the coast redwood range.

Alternatively, global atmospheric transport models may provide information on the background signal. Our previously published COS simulations using the GEOS-Chem model (Wang et al., 2016) were consistent with the large-scale temporal trends in the Sutro record (Figure 8b). The model





captures the 50 ppt increase on 16 May as well as the temporary depression on 29 and 30 May that coincided with a low pressure system that caused continental air to reach the coast. Despite the coarse spatial resolution of the global model, the similarities between Sutro and the global model suggest significant model skill in simulating synoptic variation in the background signal.

#### 4. Discussion and Conclusions

We proposed the use of COS as a new biogeochemical tracer for this coastal system to address limitations with existing ecological measurements in coast redwoods. The measurement and model results presented here are encouraging of this tracer approach for multiple reasons. First, measurement error was small relative to observed spatial and temporal variations (Figure 2). While the observed variations were approximately 50 ppt, the measurement reproducibility was less than 5 ppt for most flasks despite known challenges with water vapor spectral effects.



Figure 7. Comparison of field campaign measurements and NOAA background air-monitoring data from Sutro Tower (San Francisco, CA).



**Figure 8.** Spatial and temporal atmospheric COS trends during the CalNex period (May/June 2010). (a) Observations along the R/V *Atlantis* ship track (red) and at Sutro Tower (black). (b) Observations at Sutro Tower (black) and global GEOS-Chem simulations interpolated to the location of the Sutro Tower samples (blue). The R/V record is plotted on a different *y* axis in order to highlight the relationship in time trends between these data and the Sutro observations.

Second, the flask sampling campaign suggests that the plant uptake of COS in the redwood forest is generally consistent with our a priori knowledge. In particular, these results provide strong evidence of a close relationship between COS plant uptake, GPP, and stomatal conductance. The first line of evidence is that the daytime COS to CO<sub>2</sub> ratio in the redwoods is also observed in continental and hemispheric observations (Table 2). These large-scale correlations have already been shown to be driven by an approximate relationship between COS plant uptake and GPP and NEE (Berry et al., 2013; Campbell et al., 2008), which implies that the correlation in the redwoods has a similar driver. An additional line of evidence is that our a priori simulation of spatial variation in COS drawdown that is driven by a GPP model is similar to the spatial variation in our flask measurements (Figure 5). While model results will require further investigation of the role of transport errors (e.g., model evaluation with meteorological measurement and multiple chemical tracers), these preliminary model results suggest that the observed spatial variation is consistent with atmospheric COS variation that is driven by known plant uptake processes.

Third, the potentially confounding factors from soil fluxes, anthropogenic emissions, and nighttime uptake are relatively insignificant in this domain. While continental and global applications of the COS tracer are supported by the dominance of plant uptake and the ocean sources, ecosystem scale application can be confounded by the presence of local sources and sinks that are not associated with daytime COS plant uptake (Commane et al., 2015; Maseyk et al., 2014; Wohlfahrt et al., 2012). However, we found no evidence for significant ecosystem sources in the measurements or simu-

lations. In particular, our grassland measurements at Big Basin should be largely influenced by soil fluxes but were similar to coastal inflow measurements suggesting only small soil fluxes. Furthermore, the nocturnal correlation of COS and CO<sub>2</sub> is quantitatively consistent with previous work that shows a relatively small effect from nocturnal COS plant uptake and soils (Maseyk et al., 2014). A rough approximate of the nighttime COS flux can be obtained by combining our measured nighttime correlation of -3 ppt COS/ppm CO<sub>2</sub> with an estimated nighttime CO<sub>2</sub> ecosystem source of 3 µmol CO<sub>2</sub> m<sup>-2</sup> s<sup>-1</sup> using the Vegetation Photosynthesis and Respiration Model (VPRM) (Mahadevan et al., 2008), which yields a nighttime COS ecosystem sink that is 9 pmol m<sup>-2</sup> s<sup>-1</sup>, only approximately 20% of expected daytime COS plant uptake.

Sources and sinks of COS from fog water also need to be understood and quantified for COS analysis. Our order of magnitude analysis suggests that the influence of fog water will be small relative to plant uptake (see supporting information) (Belviso et al., 1967; Elliot et al., 1989; Mu et al., 2004). Nevertheless, measurements of these processes in redwood forests should be made in future studies to be sure.

Fourth, we found that, although variation in the background signal is significant, it can be characterized by multiple approaches. These approaches include site-level measurements at the coast, NOAA air-monitoring data from Sutro Tower, and global atmospheric chemical transport simulations.

Given the relatively slow pace of the decline in fog, it is unlikely that this measurement technique would be able to detect a response trend in the redwood stomatal conductance and GPP through short-term COS observations. Instead, the COS measurement approach would be better suited to quantify the redwood response to the much larger day-to-day variations in fog. The process knowledge obtained from such measurements could be combined with midcentury fog projections to provide a scenario analysis of potential redwood vulnerabilities.

While these results are encouraging of the biogeochemical tracer approach, they are also supportive of meteorological tracer applications. Continental outflow events observed from the R/V Atlantis consistently show 10% to 12% depletions in COS due to the widespread continental sink. In contrast to the COS signal, atmospheric CO<sub>2</sub> in these outflow events varies from a decline of 1% to growth of 4% due to the heterogeneous continental CO<sub>2</sub> sources and sinks. Thus, the large and consistent signal of COS provides a clear

indicator of continental influence, which could be used for a variety of applications including filtering out local signals in NOAA's coastal air-monitoring data (De Gouw et al., 2009; Riley et al., 2005).

The use of the COS tracer to support GPP and stomatal conductance estimation in the redwoods may also have broader applications to coastal and continental carbon budgets. North American terrestrial carbon budgets can be inferred using observations of the CO<sub>2</sub> continental inflow on the Pacific coast and outflow on the Atlantic coast. While Pacific coast observations under westerly winds provide a basis for the inflow observation, significant noise is introduced into these measurements from ecosystem sources and sinks that are adjacent to the coastal observatories (Ganguly et al., 2011; Riley et al., 2005). Significant terrestrial CO<sub>2</sub> fluxes in coastal systems have been observed by high-resolution, top-down analysis (Ahmadov et al., 2009; Göckede et al., 2010; Gourdji et al., 2011). In situ COS measurements might characterize Pacific Coast inflow with sharply reduced complications from these varied and co-located coastal CO<sub>2</sub> sources and sinks.

#### **Data Availability**

Flask observations of COS and  $CO_2$  mixing ratios are available in the supporting information.

#### References

Ahmadov, R., Gerbig, C., Kretschmer, R., Körner, S., Rödenbeck, C., Bousquet, P., & Ramonet, M. (2009). Comparing high resolution WRF-VPRM simulations and two global CO<sub>2</sub> transport models with coastal tower measurements of CO<sub>2</sub>. *Biogeosciences*, 6(5), 807–817.

Ambrose, A. R., Sillett, S. C., & Dawson, T. E. (2009). Effects of tree height on branch hydraulics, leaf structure and gas exchange in California redwoods. *Plant, Cell & Environment*, 32(7), 743–757.

Angevine, W. M., Eddington, L., Durkee, K., Fairall, C., Bianco, L., & Brioude, J. (2012). Meteorological model evaluation for CalNex 2010. Monthly Weather Review, 140(12), 3885–3906.

Archer, C. L., Jacobson, M. Z., & Ludwig, F. L. (2005). The Santa Cruz eddy. Part I: observations and statistics. Monthly Weather Review, 133(4), 767–782.

Asaf, D., Rotenberg, E., Tatarinov, F., Dicken, U., Montzka, S. A., & Yakir, D. (2013). Ecosystem photosynthesis inferred from measurements of carbonyl sulfide flux. *Nature Geoscience*, 6(3), 186–190.

Ball, J. T., Woodrow, I. E., & Berry, J. A. (1987). A model predicting stomatal conductance and its contribution to the control of photosynthesis under different environmental conditions. In J. Biggins (Ed.), *Progress in photosynthesis research* (pp. 221–224). New York: Springer.

- Belviso, S., Mihalopoulos, N., & Nguyen, B. C. (1967). The supersaturation of carbonyl sulfide (OCS) in rain waters. Atmospheric Environment, 21(6), 1363–1367.
- Berry, J., Wolf, A., Campbell, J. E., Baker, I., Blake, N., Blake, D., ... Zhu, Z. (2013). A coupled model of the global cycles of carbonyl sulfide and CO<sub>2</sub>: A possible new window on the carbon cycle. *Journal of Geophysical Research: Biogeosciences*, *118*, 842–852. https://doi.org/10.1002/jgrg.20068
- Blake, N. J., Campbell, J. E., Vay, S. A., Fuelberg, H. E., Meinardi, S., Rowland, F. S., & Blake, D. R. (2008). Carbonyl sulfide (OCS): Large scale distributions over North America during INTEX-NA and relationship with CO<sub>2</sub>. Journal of Geophysical Research, 113, D09S90. https://doi. org/10.1029/2007JD009163
- Bloem, E., Haneklaus, S., Kesselmeier, J., & Schnug, E. (2012). Sulfur fertilization and fungal infections affect the exchange of H2S and COS from agricultural crops. *Journal of Agricultural and Food Chemistry*, 60(31), 7588–7596.
- Burgess, S., & Dawson, T. (2004). The contribution of fog to the water relations of Sequoia sempervirens (D. Don): Foliar uptake and prevention of dehydration. *Plant, Cell & Environment, 27*(8), 1023–1034.
- Campbell, J. E., Berry, J. A., Seibt, U., Smith, S. J., Montzka, S. A., Launois, T., ... Laine, M. (2017). Large historical growth in global terrestrial gross primary production. *Nature*, 544(7648), 84–87.
- Campbell, J. E., Carmichael, G. R., Chai, T., Mena-Carrasco, M., Tang, Y., Blake, D. R., ... Stanier, C. O. (2008). Photosynthetic control of atmospheric carbonyl sulfide during the growing season. Science, 322(5904), 1085–1088.
- Campbell, J. E., Carmichael, G. R., Tang, Y., Chai, T., Vay, S. A., Choi, Y.-H., ... Stanier, C. O. (2007). Analysis of anthropogenic CO<sub>2</sub> signal in ICARTT using a regional chemical transport model and observed tracers. *Tellus B*, 59(2), 199–210. https://doi.org/10.1111/j.1600-0889.2006.00239.x
- Campbell, J., Kesselmeier, J., Yakir, D., Berry, J., Peylin, P., Belviso, S., . . . Chen, H. (2017). Assessing a new clue to how much carbon plants take up. *Eos, Transactions American Geophysical Union, 98.*
- Campbell, J. E., Whelan, M. E., Seibt, U., Smith, S. J., Berry, J. A., & Hilton, T. W. (2015). Atmospheric carbonyl sulfide sources from anthropogenic activity: Implications for carbon cycle constraints. *Geophysical Research Letters*, 42, 3004–3010. https://doi.org/10.1002/ 2015GL063445
- Commane, R., Herndon, S., Zahniser, M., Lerner, B., McManus, J., Munger, J., ... Wofsy, S. (2013). Carbonyl sulfide in the planetary boundary layer: Coastal and continental influences. *Journal of Geophysical Research: Atmospheres, 118,* 8001–8009. https://doi.org/10.1002/jgrd.50581
- Commane, R., Meredith, L. K., Baker, I. T., Berry, J. A., Munger, J. W., Montzka, S. A., ... Wofsy, S. C. (2015). Seasonal fluxes of carbonyl sulfide in a midlatitude forest. *Proceedings of the National Academy of Sciences of the United States of America*, 112(46), 14,162–14,167.

Dawson, T. E. (1998). Fog in the California redwood forest: Ecosystem inputs and use by plants. *Oecologia*, 117(4), 476–485. De Gouw, J., Warneke, C., Montzka, S., Holloway, J., Parrish, D., Fehsenfeld, F., ... Flocke, F. (2009). Carbonyl sulfide as an inverse tracer for

- biogenic organic carbon in gas and aerosol phases. Geophysical Research Letters, 36, L05804. https://doi.org/10.1029/2008GL036910
  Diffenbaugh, N. S., Snyder, M. A., & Sloan, L. C. (2004). Could CO<sub>2</sub>-induced land-cover feedbacks alter near-shore upwelling regimes?
  Proceedings of the National Academy of Sciences of the United States of America, 101(1), 27–32.
- Elliot, S., Lu, E., & Rowland, F. S. (1989). Rates and mechanisms for the hydrolysis of carbonyl sulfide in natural waters. Environmental Science & Technology, 23, 458–461.

Ewing, H. A., Weathers, K. C., Templer, P. H., Dawson, T. E., Firestone, M. K., Elliott, A. M., & Boukili, V. K. (2009). Fog water and ecosystem function: Heterogeneity in a California redwood forest. *Ecosystems*, 12(3), 417–433.

#### Acknowledgments

This study was supported by NSF Coastal SEES 1600109 and a UCOP/ISEECI fellowship (http://iseeci. ucnrs.org) to T. W. Hilton. The research of J. R. Stinecipher was partially supported by the UC Laboratory Fees Research Program (grant LGF-17-476795) and performed in part under the auspices of the U.S. DOE by LLNL under contract DE-AC52-07NA27344. This research used resources of the National Energy Research Scientific Computing Center (NERSC), a DOE Office of Science User Facility supported by the Office of Science of the U.S. DOE under contract DE-AC02-05CH11231. Site access and field support were provided at the University of California Natural Reserve System Landels-Hill Big Creek Reserve (M. Readdie, S. McStay, F. Arias, C. Williams, and R. Næsborg) and Big Basin State Park (J. Kerbavaz and E. Bertram). We thank W. M. Angevine for meteorology data. We thank M. L. Fischer for supporting the NOAA flask sampling at Mount Sutro through support from the California Energy **Commissions Public Interest** Environmental Research program to the Lawrence Berkelev National Laboratory under contract DE-AC02-05CH11231. Wind profiler data from the Naval Postgraduate School were provided by W. Nuss and R. Lind. Reference herein to any specific commercial product, process, or service by trade name, trademark, manufacturer, or otherwise does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States government or Lawrence Livermore National Security, LLC.

Fernández, M., Hamilton, H. H., & Kueppers, L. M. (2015). Back to the future: using historical climate variation to project near-term shifts in habitat suitable for coast redwood. *Global Change Biology*, 21(11), 4141–4152.

Ganguly, D., Dey, M., Chowdhury, C., Pattnaik, A., Sahu, B., & Jana, T. (2011). Coupled micrometeorological and biological processes on atmospheric CO<sub>2</sub> concentrations at the land-ocean boundary, NE coast of India. *Atmospheric Environment*, *45*(23), 3903–3910.

- Glatthor, N., Höpfner, M., Baker, I. T., Berry, J., Campbell, J. E., Kawa, S. R., ... von Clarmann, T. (2015). Tropical sources and sinks of carbonyl sulfide observed from space. *Geophysical Research Letters*, 42, 10,082–10,090. https://doi.org/10.1002/2015GL066293
- Göckede, M., Turner, D. P., Michalak, A. M., Vickers, D., & Law, B. E. (2010). Sensitivity of a subregional scale atmospheric inverse CO<sub>2</sub> modeling framework to boundary conditions. *Journal of Geophysical Research*, *115*, D24112. https://doi.org/10.1029/2010JD014443
- Gourdji, S. M., Mueller, K. L., Yadav, V., Huntzinger, D. N., Andrews, A. E., Trudeau, M., ... Wen, D. (2011). North American CO<sub>2</sub> exchange: intercomparison of modeled estimates with results from a fine-scale atmospheric inversion. *Biogeosciences Discussions*, 8(4), 6775–832.
- Guo, H., Simpson, I. J., Ding, A. J., Wang, T., Saunders, S. M., Wang, T. J., ... Blake, D. R. (2010). Carbonyl sulfide, dimethyl sulfide and carbon disulfide in the Pearl River Delta of southern China: Impact of anthropogenic and biogenic sources. *Atmospheric Environment*, 44(31), 3805–3813. https://doi.org/10.1016/j.atmosenv.2010.06.040
- Hilton, T. W., Zumkehr, A., Kulkarni, S., Berry, J., Whelan, M. E., & Campbell, J. E. (2015). Large variability in ecosystem models explains uncertainty in a critical parameter for quantifying GPP with carbonyl sulfide. *Tellus B*, 67(1), 26329. https://doi.org/10.3402/tellusb.v3467.26329
- Hilton, T. W., Zumkehr, A., Kulkarni, S., Berry, J., Whelan, M. E., & Campbell, J. E. (2017). Peak growing season gross uptake of carbon in North America is largest in the Midwest USA, Naure Climate Change, 2nd round review minor revisions.
- Huntzinger, D. N., Post, W. M., Wei, Y., Michalak, A. M., West, T. O., Jacobson, A. R., ... Hoffman, F. M. (2012). North American Carbon Program (NACP) regional interim synthesis: Terrestrial biospheric model intercomparison. *Ecological Modelling*, 232, 144–157.
- Johnstone, J. A., & Dawson, T. E. (2010). Climatic context and ecological implications of summer fog decline in the coast redwood region. Proceedings of the National Academy of Sciences of the United States of America, 107(10), 4533–4538.
- Johnstone, J. A., Roden, J. S., & Dawson, T. E. (2013). Oxygen and carbon stable isotopes in coast redwood tree rings respond to spring and summer climate signals. *Journal of Geophysical Research: Biogeosciences, 118*, 1438–1450. https://doi.org/10.1002/jgrg.20111
- Kettle, A. J., Kuhn, U., von Hobe, M., Kesselmeier, J., & Andreae, M. O. (2002). Global budget of atmospheric carbonyl sulfide: Temporal and spatial variations of the dominant sources and sinks. *Journal of Geophysical Research*, 107(D22), 4658. https://doi.org/10.1029/ 2002JD002187
- Knyazikhin, Y., Glassy, J., Privette, J., Tian, Y., Lotsch, A., Zhang, Y., ... Myneni, R. (1999). MODIS leaf area index (LAI) and fraction of photosynthetically active radiation absorbed by vegetation (FPAR) product (MOD15) algorithm theoretical basis document, Theoretical Basis Document, NASA Goddard Space Flight Center, Greenbelt, MD, 20771.
- Kooijmans, L. M. J., Maseyk, K., Seibt, U., Sun, W., Vesala, T., Mammarella, I., & Kolari, P. (2017). Canopy uptake dominates nighttime carbonyl sulfide fluxes in a boreal forest. Atmospheric Chemistry and Physics, 17, 11,453–11,465.
- Kooijmans, L. M. J., Uitslag, N. A. M., Zahniser, M. S., Nelson, D. D., Montzka, S. A., & Chen, H. (2016). Continuous and high-precision atmospheric concentration measurements of COS, CO<sub>2</sub>, CO and H2O using a quantum cascade laser spectrometer (QCLS). Atmospheric Measurement Techniques, 9(11), 5293–5314.
- Kuai, L., Worden, J. R., Campbell, J. E., Kulawik, S. S., Li, K. F., Lee, M., ... Berry, J. A. (2015). Estimate of carbonyl sulfide tropical oceanic surface fluxes using Aura Tropospheric Emission Spectrometer observations. *Journal of Geophysical Research: Atmospheres*, 120, 11,012–11,023. https://doi.org/10.1002/2015JD023493
- Launois, T., Belviso, S., Bopp, L., Fichot, C. G., & Peylin, P. (2015). A new model for the global biogeochemical cycle of carbonyl sulfide—Part 1: Assessment of direct marine emissions with an oceanic general circulation and biogeochemistry model. *Atmospheric Chemistry and Physics*, *15*(5), 2295–2312.
- Li, X., Zheng, W., Pichel, W. G., Zou, C.-Z., & Clemente-Colón, P. (2007). Coastal katabatic winds imaged by SAR. *Geophysical Research Letters*, 34, L03804. https://doi.org/10.1029/2006GL028055
- Limm, E. B., Simonin, K. A., Bothman, A. G., & Dawson, T. E. (2009). Foliar water uptake: A common water acquisition strategy for plants of the redwood forest. *Oecologia*, 161(3), 449–459.
- Mahadevan, P., Wofsy, S. C., Matross, D. M., Xiao, X., Dunn, A. L., Lin, J. C., ... Gottlieb, E. W. (2008). A satellite-based biosphere parameterization for net ecosystem CO<sub>2</sub> exchange: Vegetation Photosynthesis and Respiration Model (VPRM). *Global Biogeochemical Cycles*, 22, B2005. https://doi.org/10.1029/2006GB002735
- Maseyk, K., Berry, J. A., Billesbach, D., Campbell, J. E., Torn, M. S., Zahniser, M., & Seibt, U. (2014). Sources and sinks of carbonyl sulfide in an agricultural field in the Southern Great Plains. Proceedings of the National Academy of Sciences of the United States of America, 111(25), 9064–9069.
- Montzka, S. A., Hall, B., Elkins, J. W., Miller, L., Watson, A., Sweeney, C., & Tans, P. P. (2007). On the global distribution, seasonality, and budget of atmospheric carbonyl sulfide (COS) and some similarities to CO<sub>2</sub>. Journal of Geophysical Research, 112, D09302. https://doi.org/10.1029/ 2006JD007665
- Mu, Y., Geng, C., Wang, M., Wu, H., Zhang, X., & Jiang, G. (2004). Photochemical production of carbonyl sulfide in precipitation. Journal of Geophysical Research, 109, D13301. https://doi.org/10.1029/2003JD004206
- Potter, C. (2012). Net primary production and carbon cycling in coast redwood forests of central California. *Open Journal of Ecology*, 2(3), 147–153. https://doi.org/10.4236/oje.2012.23018
- Riley, W., Randerson, J., Foster, P., & Lueker, T. (2005). Influence of terrestrial ecosystems and topography on coastal CO<sub>2</sub> measurements: A case study at Trinidad Head, California. *Journal of Geophysical Research*, *110*, G01005. https://doi.org/10.1029/2004JG000007
- Sandoval-Soto, L., Stanimirov, M., von Hobe, M., Schmitt, V., Valdes, J., Wild, A., & Kesselmeier, J. (2005). Global uptake of carbonyl sulfide (COS) by terrestrial vegetation: Estimates corrected by deposition velocities normalized to the uptake of carbon dioxide (CO<sub>2</sub>). *Biogeosciences*, 2(2), 125–132.
- Sandu, A., Daescu, D. N., Carmichael, G. R., & Chai, T. (2005). Adjoint sensitivity analysis of regional air quality models. *Journal of Computational Physics*, 204(1), 222.
- Sawaske, S. R., & Freyberg, D. L. (2015). Fog, fog drip, and streamflow in the Santa Cruz Mountains of the California Coast Range. *Ecohydrology*, 8(4), 695–713.
- Seibt, U., Kesselmeier, J., Sandoval-Soto, L., Kuhn, U., & Berry, J. A. (2010). A kinetic analysis of leaf uptake of COS and its relation to transpiration, photosynthesis and carbon isotope fractionation. *Biogeosciences*, 7(1), 333–341.
- Simonin, K. A., Santiago, L. S., & Dawson, T. E. (2009). Fog interception by Sequoia sempervirens (D. Don) crowns decouples physiology from soil water deficit. *Plant, Cell & Environment*, 32(7), 882–892.
- Stimler, K., Berry, J. A., & Yakir, D. (2012). Effects of carbonyl sulfide and carbonic anhydrase on stomatal conductance. *Plant Physiology*, 158(1), 524–530.

Stimler, K., Montzka, S. A., Berry, J. A., Rudich, Y., & Yakir, D. (2010). Relationships between carbonyl sulfide (COS) and CO<sub>2</sub> during leaf gas exchange. *The New Phytologist*, 186(4), 869–878.

Stimler, K., Nelson, D., & Yakir, D. A. N. (2009). High precision measurements of atmospheric concentrations and plant exchange rates of carbonyl sulfide using mid-IR quantum cascade laser. *Global Change Biology*, 16(9), 2496–2503.

Sun, W., Maseyk, K., Lett, C., & Seibt, U. (2016). Litter dominates surface fluxes of carbonyl sulfide (COS) in a Californian oak woodland. Journal of Geophysical Research: Biogeosciences, 121, 438–450. https://doi.org/10.1002/2015JG003149

Tseng, Y.-H., Chien, S.-H., Jin, J., & Miller, N. L. (2012). Modeling air-land-sea interactions using the Integrated Regional Model System in Monterey Bay, California. *Monthly Weather Review*, 140(4), 1285–1306.

Vacher, C., Hampe, A., Porté, A. J., Sauer, U., Compant, S., & Morris, C. E. (2016). The phyllosphere: Microbial jungle at the plant-climate interface. *Annual Review of Ecology, Evolution, and Systematics*, 47, 1–24.

Wang, Y., Deutscher, N. M., Palm, M., Warneke, T., Notholt, J., Baker, I., ... Kremser, S. (2016). Towards understanding the variability in biospheric CO<sub>2</sub> fluxes: Using FTIR spectrometry and a chemical transport model to investigate the sources and sinks of carbonyl sulfide and its link to CO<sub>2</sub>. Atmospheric Chemistry and Physics, 16(4), 2123–2138. https://doi.org/10.5194/acp-16-2123-2016

Wehr, R., Commane, R., Munger, J. W., McManus, J. B., Nelson, D. D., Zahniser, M. S., ... Wofsy, S. C. (2017). Dynamics of canopy stomatal conductance, transpiration, and evaporation in a temperate deciduous forest, validated by carbonyl sulfide uptake. *Biogeosciences*, 14(2), 389.

Whelan, M. E., Hilton, T. W., Berry, J. A., Berkelhammer, M., Desai, A. R., & Campbell, J. E. (2016). Carbonyl sulfide exchange in soils for better estimates of ecosystem carbon uptake. *Atmospheric Chemistry and Physics*, 16(6), 3711–3726.

Whelan, M. E., & Rhew, R. C. (2016). Reduced sulfur trace gas exchange between a seasonally dry grassland and the atmosphere. *Biogeochemistry*, 128(3), 267–280.

Wohlfahrt, G., Brilli, F., Hörtnagl, L., Xu, X., Bingemer, H., Hansel, A., & Loreto, F. (2012). Carbonyl sulfide (COS) as a tracer for canopy photosynthesis, transpiration and stomatal conductance: Potential and limitations. *Plant, Cell & Environment, 35*(4), 657–667.

Zumkehr, A., Hilton, T. W., Whelan, M., Smith, S., & Campbell, J. E. (2017). Gridded anthropogenic emissions inventory and atmospheric transport of carbonyl sulfide in the U.S. *Journal of Geophysical Research: Atmospheres*, 122, 2169–2178. https://doi.org/10.1002/ 2016JD025550